

Citation: Catley D, Puoane T, Tsolekile L, Resnicow K, Fleming KK, Hurley EA, et al. (2022) Evaluation of an adapted version of the Diabetes Prevention Program for low- and middle-income countries: A cluster randomized trial to evaluate "Lifestyle Africa" in South Africa. PLoS Med 19(4): e1003964. https://doi.org/10.1371/journal.pmed.1003964

Academic Editor: Weiping Jia, Shanghai Jiao Tong University Affiliated Sixth People's Hospital, CHINA

Received: September 2, 2021

Accepted: March 14, 2022

Published: April 15, 2022

Copyright: © 2022 Catley et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The necessary inclusion of some information in the data set (i.e., participant age, NGO, phone number, dates of participation) creates the possibility of deductive disclosure for participants. Similar risk could exist for other unique characteristics or a unique combination of characteristics. Therefore, we will make data collected available only upon request from users who can show proof of human subject's training and only under a data-sharing RESEARCH ARTICLE

Evaluation of an adapted version of the Diabetes Prevention Program for low- and middle-income countries: A cluster randomized trial to evaluate "Lifestyle Africa" in South Africa

Delwyn Catley^{1,2}*, Thandi Puoane³, Lungiswa Tsolekile³, Ken Resnicow⁴, Kandace K. Fleming⁵, Emily A. Hurley^{2,6}, Joshua M. Smyth⁷, Frank T. Materia⁶, Estelle V. Lambert⁸, Mara Z. Vitolins⁹, Naomi S. Levitt¹⁰, Kathy Goggin^{2,6,11}

 Center for Children's Healthy Lifestyles and Nutrition, Children's Mercy Kansas City, Kansas City, Missouri, United States of America, 2 University of Missouri-Kansas City School of Medicine, Kansas City, Missouri, United States of America, 3 University of the Western Cape School of Public Health, Cape Town, South Africa, 4 University of Michigan School of Public Health, Ann Arbor, Michigan, United States of America,
Life Span Institute, University of Kansas, Lawrence, Kansas, United States of America, 6 Health Services and Outcomes Research, Children's Mercy Hospitals and Clinics, Kansas City, Missouri, United States of America, 7 College of Health and Human Development, Penn State University, University Park, Pennsylvania, United States of America, 8 UCT Research Centre for Health through Physical Activity, Lifestyle and Sport (HPALS), Division of Research Unit for Exercise Science and Sports Medicine, Faculty of Health Sciences, University of Cape Town, Cape Town, South Africa, 9 Department of Epidemiology & Prevention, Wake Forest School of Medicine, Winston-Salem, North Carolina, United States of America, 10 Department of Medicine and Chronic Disease Initiative for Africa, Faculty of Health Sciences, University of Cape Town, Cape Town, South Africa, 11 University of Missouri-Kansas City School of Pharmacy, Kansas City, Missouri, United States of America

* dcatley@cmh.edu

Abstract

Background

Low- and middle-income countries (LMICs) are experiencing major increases in diabetes and cardiovascular conditions linked to overweight and obesity. Lifestyle interventions such as the United States National Diabetes Prevention Program (DPP) developed in high-income countries require adaptation and cultural tailoring for LMICs. The objective of this study was to evaluate the efficacy of "Lifestyle Africa," an adapted version of the DPP tailored for an underresourced community in South Africa compared to usual care.

Methods and findings

Participants were residents of a predominantly Xhosa-speaking urban township of Cape Town, South Africa characterized by high rates of poverty. Participants with body mass index (BMI) \geq 25 kg/m² who were members of existing social support groups or "clubs" receiving health services from local nongovernmental organizations (NGOs) were enrolled in a cluster randomized controlled trial that compared Lifestyle Africa (the intervention

agreement that provides for: (1) use of the data only for research purposes, (2) exclusion of any identifying or potentially identifying information in shared analyses, publications, reports, etc., (3) appropriate storage and securing of the data to prevent authorized persons from accessing it, (4) a commitment to destroy or return the data after analyses are completed. Inquiries can be sent to institutional administrators at the following email address: CRI-inquiry@cmh.edu.

Funding: This work was funded by the National Heart Lung and Blood Institute of the National Institutes of Health (R01HL126099) to DC. <u>https://</u> www.nhlbi.nih.gov/ The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: BMI, body mass index; CHW, community health worker; CI, confidence interval; CONSORT, Consolidated Standards of Reporting Trials; CVD, cardiovascular disease; DPP, Diabetes Prevention Program; HbA1c, hemoglobin A1c; HDL, high-density lipoprotein; ICC, intracluster correlation coefficient; LDL, low-density lipoprotein; LMIC, low- and middle-income country; MI, motivational interviewing; NGO, nongovernmental organization; OR, odds ratio; T2DM, type 2 diabetes mellitus. condition) to usual care (the control condition). The Lifestyle Africa intervention consisted of 17 video-based group sessions delivered by trained community health workers (CHWs). Clusters were randomized using a numbered list of the CHWs and their assigned clubs based on a computer-based random allocation scheme. CHWs, participants, and research team members could not be blinded to condition. Percentage weight loss (primary outcome), hemoglobin A1c (HbA1c), blood pressure, triglycerides, and low-density lipoprotein (LDL) cholesterol were assessed 7 to 9 months after enrollment. An individual-level intention-to-treat analysis was conducted adjusting for clustering within clubs and baseline values. Trial registration is at ClinicalTrials.gov (NCT03342274). Between February 2018 and May 2019, 782 individuals were screened, and 494 were enrolled. Participants were predominantly retired (57% were receiving a pension) and female (89%) with a mean age of 68 years. Participants from 28 clusters were allocated to Lifestyle Africa (15, n = 240) or usual care (13, n = 254). Fidelity assessments indicated that the intervention was generally delivered as intended. The modal number of sessions held across all clubs was 17, and the mean attendance of participants across all sessions was 61%. Outcome assessment was completed by 215 (90%) intervention and 223 (88%) control participants. Intent-to-treat analyses utilizing multilevel modeling included all randomized participants. Mean weight change (primary outcome) was -0.61% (95% confidence interval (CI) = -1.22, -0.01) in Lifestyle Africa and -0.44% (95% CI = -1.06, 0.18) in control with no significant difference (group difference = -0.17%; 95% CI = -1.04, 0.71; p = 0.71). However, HbA1c was significantly lower at follow-up in Lifestyle Africa compared to the usual care group (mean difference = -0.24, 95% CI = -0.39, -0.09, p = 0.001). None of the other secondary outcomes differed at follow-up: systolic blood pressure (group difference = -1.36; 95% CI = -6.92, 4.21; p = 0.63), diastolic blood pressure (group difference = -0.39; 95% CI = -3.25, 2.30; p= 0.78), LDL (group difference = -0.07; 95% CI = -0.19, 0.05; p = 0.26), triglycerides (group difference = -0.02; 95% CI = -0.20, 0.16; p = 0.80). There were no unanticipated problems and serious adverse events were rare, unrelated to the intervention, and similar across groups (11 in Lifestyle Africa versus 13 in usual care). Limitations of the study include the lack of a rigorous dietary intake measure and the high representation of older women.

Conclusions

In this study, we found that Lifestyle Africa was feasible for CHWs to deliver and, although it had no effect on the primary outcome of weight loss or secondary outcomes of blood pressure or triglycerides, it had an apparent small significant effect on HbA1c. The study demonstrates the potential feasibility of CHWs to deliver a program without expert involvement by utilizing video-based sessions. The intervention may hold promise for addressing cardiovascular disease (CVD) and diabetes at scale in LMICs.

Trial registration

ClinicalTrials.gov NCT03342274.

Author summary

Why was this study done?

- Low- and middle-income countries (LMICs) are experiencing high rates of chronic diseases such as diabetes and cardiovascular conditions linked to overweight and obesity.
- "Lifestyle" programs developed in high-income countries that help participants to lose weight are effective at preventing and reducing chronic disease but require adaptation to be feasible and effective in LMICs.

What did the researchers do and find?

- This study evaluated whether a lifestyle program (Lifestyle Africa) adapted for overweight participants in an underresourced community in South Africa could improve upon usual care in reducing participants' weight and improving other indicators of health.
- Overweight volunteers from 28 preexisting "health clubs" receiving health services from community health workers (CHWs) were randomly allocated to Lifestyle Africa (15 clubs, n = 240) or usual care (13 clubs, n = 254).
- Participants in the Lifestyle Africa program participated in regular group sessions to learn strategies for weight loss through healthy eating and physical activity.
- Participants' weight, blood pressure, cholesterol, and hemoglobin A1c (HbA1c; an indicator of mean blood sugar over 3 months and related to diabetes status) were assessed at the beginning of the trial and 7 to 9 months later.
- Results revealed no effect of the intervention on the main outcome of weight loss (mean weight reduction was just 0.17% greater in Lifestyle Africa), blood pressure, or cholesterol. However, the intervention resulted in lower HbA1c (the mean was 0.24 lower in Lifestyle Africa).

What do these findings mean?

- The results indicate that Lifestyle Africa was feasible for CHWs to deliver, and the significant effect on HbA1c suggests that the intervention may have potential for addressing chronic disease at scale in LMICs.
- However, the relatively small effect on HbA1c and lack of effect on other outcomes indicate that the intervention needs to be strengthened before being recommended for widescale use.
- The study was limited by lack of an accurate measure of any dietary changes among participants as well as the high number of older females who may have respond in a unique way to the intervention.

Introduction

Approximately 15 million premature deaths per year worldwide are attributable to noncommunicable diseases, including cardiovascular diseases (CVDs) and diabetes mellitus [1]. Over 85% of these deaths occur in low- and middle-income countries (LMICs), where the disproportionate burden of noncommunicable disease is expected to increase [2]. In South Africa, noncommunicable diseases are more prevalent than HIV/AIDS and tuberculosis combined, with CVD being the leading category [3]. The prevalence of type 2 diabetes mellitus (T2DM) in South Africa is estimated to be about 15% with higher rates among women and older people [4].

The increase in CVD and T2DM in LMICs is attributed partly to rising incomes and urbanization, which leads to a shift from eating unrefined carbohydrates to greater intake of fats, sweeteners, and animal source foods, as well as highly processed foods [5]. Urbanization is also linked to reductions in physical activity [6]. Cultural beliefs and practices (e.g., less stigmatization of obesity and a preference for being overweight) may also play a role [7].

In high-income countries, lifestyle interventions such as the US Diabetes Prevention Program (DPP) have been developed to combat unhealthy changes in diet and physical activity [8]. In the program "lifestyle coaches," who are usually health professionals such dietitians, nutritionists, or counselors, encourage participants to engage in at least 150 minutes of moderate physical activity per week and to reduce body weight by 7% over 6 months. Randomized trials have demonstrated powerful clinical effects in preventing the development of T2DM [8,9] as well as reducing weight and improving glucose control, blood pressure, triglycerides, and high-density lipoprotein (HDL) cholesterol among individuals with T2DM [10].

Similar lifestyle programs targeting weight change through both physical activity and diet have also been evaluated in randomized trials in some lower income countries including in Thailand [11], Brazil [12], and China [13], with generally positive health effects [14]; however, these interventions were delivered by researchers or high-level health professionals and mostly through health clinics or work sites. The reach of lifestyle interventions in LMICs will be greatly limited if they depend on scarce and expensive professional healthcare workers and if participant time and travel costs are barriers to access. Studies in the US have shown that lay or community health workers (CHWs) can be successfully deployed as DPP interventionists [15], but randomized trials in LMICs are lacking. Two notable exceptions are the Kerala DPP in India [16] and the Grenada Heart Project [17], both of which used a peer support model of intervention. Both studies found some positive effects on composite health risk indicators, but neither study found significant effects on key clinical indicators such as weight, hemoglobin A1c (HbA1c), or blood pressure.

To extend this work, we developed a new version of the DPP ("Lifestyle Africa") tailored specifically for an underresourced community in South Africa and designed to be delivered by CHWs. South Africa has a long history of employing CHWs as key interventionists in its national primary care program where they connect community members with services provided in local health facilities [18]. The purpose of our study was to test the adapted program in a cluster randomized trial with CHW teams and their associated clubs as clusters. The primary outcome analysis compared percentage weight loss between Lifestyle Africa and usual care participants. HbA1c, blood pressure, triglycerides, and low-density lipoprotein (LDL) cholesterol were selected as secondary outcomes because prior studies of lifestyle interventions have more typically reported changes in weight [9,19].

Methods and analysis

Methods

Study design and participants. This study was a 2-arm, parallel group, cluster-designed randomized controlled trial with a planned follow-up and crossover of the control arm after 1 year. Participants were enrolled in 2 waves separated by a year. Although the goal was to conduct a 1-year follow-up, we chose to avoid conducting our assessment after the long annual December to January holiday break in which most community members and CHWs travel to rural areas. We believed the disruption of routine and session attendance had the potential to confound our assessment. Enrollment and baseline assessment therefore took place as early as possible in the calendar year (February and March), and the main outcome assessment time point was as late as possible in the same calendar year (i.e., mid-September through early December) resulting in follow-up taking place approximately 7 to 9 months after enrollment. Mean, median, and interguartile range of follow-up was 210, 212, and 28.5 days for the control group and 224, 224, and 21.5 days for the Lifestyle Africa group. This report focuses on our primary and secondary biologic outcomes at the first follow-up. Full details of the study are described elsewhere [20,21]. The study was approved by the ethics committees at the University of Cape Town (primary; number 109/2017) and Children's Mercy Kansas City (number 15080328) and is registered with Clinical-Trials.gov, number NCT03342274. This study is reported per the Consolidated Standards of Reporting Trials (CONSORT) guidelines (S1 Checklist).

The study setting was a predominantly Xhosa-speaking urban township of Cape Town, South Africa characterized by high rates of poverty and unemployment, low education, and high rates of overweight and obesity [22,23]. Our partner NGOs were utilizing CHWs to support delivery of basic healthcare services (e.g., medication delivery and wellness programs), and CHWs had received basic training in in-home–based care, chronic disease management, and wellness. In consultation with their CHWs, our partner NGOs adopted the Lifestyle Africa Program as their standard of care (training and oversight described below). CHWs varied in their educational attainment but had sufficient reading, writing, and arithmetic skills to maintain attendance registers and medication logs. CHW services were being provided through community clubs that met at community facilities (e.g., churches) or homes.

To reach our recruitment goal, we planned and conducted recruitment at all clubs in one branch of one partner NGO and all clubs from the other partner NGO. Two scripted video-based recruitment sessions were held at club meetings to introduce the study to members. Recruitment sessions were led by either CHWs or research team members. Members were invited to attend a club enrollment session held at a convenient location. At the enrollment session, participants completed written informed consent, eligibility screening, and baseline assessment. To serve as many club members as possible, that criterion for inclusion was limited to being overweight or obese (i.e., $BMI \ge 25 \text{ kg/m}^2$). There was no exclusion related to diabetes diagnosis. If a club member's BMI could not be established because a physical disability prevented the assessment of weight the individual was not included. Exclusion criteria were the following: (1) having a blood pressure >160 (systolic) and/or >100 mm Hg (diastolic) [24]; (2) very high blood sugar (HbA1c > 11%) [19]; (3) pregnant, breastfeeding, or planning to become pregnant; (4) use of oral steroid medications that may affect weight loss; (5) not planning to stay in the club over the next 2 years; and/or (6) exhibiting intellectual disabilities that would prevent necessary understanding of the program requirements.

Randomization and masking. Clusters constituted "CHW teams" and their associated clubs because CHWs worked individually, in pairs, and occasionally in trios with a particular club. CHW teams were stratified within NGOs and allocated using a 1:1 scheme to Lifestyle Africa or usual care. Clusters were randomized by the statistician (KF) using a numbered list

of the CHWs and their assigned clubs based on a computer-based random number generated allocation scheme. Clusters were randomized prior to participant enrollment (and hence prior to individual consent) in order to establish which CHWs needed to be trained to deliver the intervention. The nature of the trial precluded blinding of CHWs, study staff, or participants.

Lifestyle Africa intervention. The Lifestyle Africa intervention was adapted from the US Centers for Disease Control National DPP [25]. The US DPP is a group-based behavioral intervention program to encourage physical activity and dietary changes that will lead to clinically meaningful weight loss. Session content is broadly based on social cognitive and problem-solving theory principles of behavior change and encompass topics such as goal setting, self-monitoring of physical activity of diet, healthy nutrition, stress management, and stimulus control [26]. During group sessions, participants receive informational content as well as mutual support through the opportunity to discuss their individual challenges and successes. The program is typically delivered by health professionals with training in nutrition or counseling. Preliminary work involved formative research conducted in collaboration with 2 community advisory boards [27]. The community advisory boards informed the development of the program and the design of the training manual and curriculum. Program development also included demonstration of session feasibility in pilot testing [28].

The Lifestyle Africa intervention retained the DPP goals for participants of 7% of weight loss and 150 min/week of physical activity and key elements from the 16 "core" and 15 "post-core" DPP sessions [25,26], while integrating necessary cultural and language adaptations relevant for the local community. For example, intervention content was adapted to explain the connection between chronic disease and lifestyle in accessible language. Culturally and liter-acy-adapted methods for calculating the number of physical activity minutes and the number of calories consumed were devised. The number of core sessions was also increased to 17 to reduce the amount of information covered in some of the sessions.

The primary Lifestyle Africa adaptation was to allow the program to be facilitated by CHWs rather than highly trained clinician facilitators with expert knowledge related to diet, physical activity, and behavior change. To make this feasible, CHWs were trained to provide participants with core intervention content via a video-based curriculum that closely followed the session content of the original DPP. To cover the content without excessive session length, an additional session was created, resulting in 17 rather than 16 sessions. Lifestyle Africa session videos were developed in the Xhosa language and featured a first-language Xhosa-speaking narrator in conjunction with animation and photos. Pauses were built into the videos when CHWs engaged participants with questions and activities (e.g., completing worksheets that personalized and reinforced concepts). Participants received a program book in either Xhosa or English (if preferred) with worksheets, information, and forms needed for each program session (e.g., goal setting sheets). CHWs used a facilitator manual in which each Lifestyle Africa session video had an accompanying "session guide" with step-by-step instructions and scripts for facilitating each of the 17 core program sessions. In addition, 12 similar but briefer "postcore" sessions were developed without video content. CHWs delivered the program weekly (or sometimes biweekly to accommodate a club's meeting schedule), although necessary adjustments were made to allow for circumstances when clubs could not meet (e.g., holidays, pension collection days, and neighborhood protests). Clubs that completed the 17 core sessions continued with monthly postcore meetings until the follow-up assessment.

To enhance the efficacy of the program, text messaging (i.e., SMS) was employed to promote healthy lifestyle modifications and behavior change [29]. A text messaging system sent 2 texts per day (late morning and evening) to participants' mobile phones and provided program reminders, fostered self-efficacy and motivation, affirmed behavior change efforts, and aided in implementation planning (e.g., healthy lifestyle tips such as "changing the way you stay active will help keep exercising fun"). The same texts were delivered to all participants, with weekly message content adapted over time to refer to the content of the session the participant had most recently attended.

CHW training included 3 days of didactic interactive workshops addressing group facilitation, diabetes management, behavior change theory, and motivational interviewing (MI). This was followed by 8 weekly half-day sessions of experiential training as mock Lifestyle Africa participants in which they participated in every core session and took turns practicing all key facilitation activities and skills (e.g., conducting the weighing in of participants, setting up the projector and using the video, and facilitating the opening discussion of the session). These trainings were provided by South African research team members in the Xhosa language (including LT and TP), who were themselves trained by authors DC, KG, and KR.

CHWs were supervised by NGO managers in accord with their usual practice. Assistance was provided as needed by research team members who observed a subset of Lifestyle Africa sessions to ensure intervention fidelity. Research team members observed the first 10 sessions for each club and then about 1 session every 5 to 8 weeks. Team members used checklists to verify the extent of completion (not completed, partly completed, and completed) of specific session facilitation elements (e.g., set up and played the session videos successfully, completed the weigh-in, and followed the steps in the session script). They also completed an adapted version of the OnePass measure for MI competence [30] to rate the quality of group facilitation, MI skills, and overall performance on a 7-point Likert scale (1 = poor/never, 4 = good/often; and 7 = excellent/always). Ratings of 4 or higher were considered acceptable.

Usual care. In usual care, CHWs continued to lead clubs in their usual activities of approximately monthly monitoring of weight, blood pressure, and blood glucose and delivery of medication. These clinical assessments were not used in the study.

Enrollment and assessment. Enrollment and baseline assessment began with wave 1 in 2018. Follow-up assessments were held at a convenient location (e.g., the club facility or a nearby community hall) for club members and were conducted in the same fashion as the baseline assessment. If necessary, transportation to the assessment site was provided. A flyer with instructions to wear light clothing, to use the restroom, and to bring medications and their mobile phone number was provided before each assessment session. After completing written informed consent participants proceeded through each assessment station including the biometric assessments and interviewer-administered survey. Participants received a R150 (approximately \$12 USD) gift voucher for completing each assessment.

Outcomes

The primary outcome was percentage of weight lost between the baseline and the follow-up visit, measured at the individual level. Weight was measured to the half kilogram with a portable electronic platform scale (Seca 876) after removal of shoes and outer clothing. Height was measured at baseline for purposes of calculating body mass index (BMI), which was calculated as weight in kilograms divided by the square of height in meters. Secondary outcomes were HbA1c, nonfasting LDL cholesterol, triglycerides, and blood pressure. HbA1c and lipids were all measured via an automated capillary sample assay using an Afinion AS100 analyzer (Alere, Waltham, Massachusetts, USA), which has shown very good agreement with laboratory methods [31,32]. Blood pressure was determined with an Omron HBP1300 instrument (Omron Healthcare, Kyoto, Kyoto, Japan) [33]. Two measurements were obtained and, if either systolic or diastolic blood pressure measurement were discrepant by more than 5 mm Hg, a third

measurement was taken. For analysis, the mean of the last 2 measurements taken was used. Participants brought all of their medications to the enrollment session to determine medication usage, the names of which were recorded by an interviewer. Demographic information including age, sex, approximate monthly income (from "no income" to "R204,801 or more" using increments that doubled at each step starting at R400-R800, R800-R1600, R1600-R3200, etc.), education (highest grade level reached from no formal schooling through any tertiary education), and relationship status (single; married; not married living with partner; separated; divorced; and widowed) was collected via interview and responses entered into a REDCap survey [34].

To monitor the trial and unexpected or severe adverse events, a Data Safety Monitoring Board oversaw the study and approved stopping rules related to serious adverse events and failure to recruit and retain participants. The board comprised members with expertise in public health, medicine, psychology, and statistics and operated independently from the study team and the funder. The board reviewed serious adverse events as well as enrollment, session completion, and follow-up rates in twice yearly meetings. Only serious adverse events and unanticipated problems were monitored due to the low-risk nature of the trial. In closed session, the board made decisions regarding continuation of the trial.

Statistical analysis

Power analysis was based on an anticipated sample of 28 total clusters averaging 18 to 19 participants for a total N of 518. We assumed a treatment weight loss of $7.21\% \pm 0.57\%$ (-7.1 kg) for intervention participants and 1% (<1.0 kg) weight loss for controls [35], yielding a conservative estimated effect size for weight loss of 0.37. Lindström and colleagues [36] reported reduction in HbA1c with an effect size of 0.31. Power analyses indicated that a trial with 28 randomly assigned clubs and 20 participants per club would provide >80% power to detect effect sizes of 0.27 with an intracluster correlation of 0.01 and 0.34 with an intracluster correlation of 0.05. Allowing for 25% attrition at the participant level (i.e., assuming 15 participants per cluster) and assuming intracluster correlation coefficients (ICC) between 0.01 and 0.05, we were adequately powered to detect effects between 0.31 and 0.37.

Our prespecified analysis plan is outlined in our published study protocol [20]. Subsequent to publication of the protocol the study sponsor (the National Institutes of Health) required the addition of analysis of outcomes by sex to our analysis plan. In accord with our prespecified plan, preliminary analyses examined baseline equivalence across arms to identify covariates for the main analyses. Prespecified multilevel modeling analyses were conducted using the MIXED procedure within SAS 9.4 (SAS Institute, Cary, North Carolina, USA). ICCs were calculated from empty models with no predictors. Initially, we examined intent-to-treat models that included occasions (baseline/year 1) nested within participants (Level 2), which were nested within clusters (Level 3). Planned examination of simple effects of condition at baseline revealed that there were not statistically significant differences between groups for any of our outcomes of interest. However, some outcomes had relatively small *p*-values (e.g., weight, p = 0.10), suggesting that analyses examining group differences at follow-up controlling for baseline levels would be appropriate. Thus, post hoc analyses examined models where participants (Level 1 units) were nested within clusters (Level 2 units) with outcomes at the end of year 1 as the dependent variable and baseline values as covariates.

For the primary outcome, percentage weight loss from baseline to the end of year 1 was the dependent variable in the multilevel models described above. Random intercepts were modeled across clusters. Similar models were evaluated for each of the secondary outcomes (blood pressure, HbA1c, LDL, and triglycerides) with outcomes at follow-up as the dependent variable and baseline level of the outcome as a covariate in each model. The first set of models included no covariates and only examined the effect of intervention. To explore any potential influence of participants baseline weight on outcomes, a second set of models was conducted, adjusting for baseline BMI. To explore the potential impact of medications we also reanalyzed any significant effects after removing participants on medications who improved in HbA1c. A third set of models were used for preplanned examination of the effect of intervention within sex groups by adding sex and sex by condition interactions to the model. Standardized effect sizes (Hedges' *g*) were calculated using differences in the model adjusted means between the treatment arms divided by the pooled observed standard deviation at baseline. To better elucidate clinical effects, we conducted post hoc analyses for significant effects (p < 0.05) using established clinical cut points. We categorized participants according to whether they lowered (improved), maintained, or raised in clinical category from baseline and examined treatment arm differences utilizing multilevel logistic regression for clustered outcomes using SAS Proc GLIMMIX with a multinomial distribution and a cumulative logit link function.

In response to peer review, we conducted additional post hoc analyses. We examined potential bias related to loss to follow-up (i.e., participants for whom follow-up data could not be collected because participants died, left the area, or could not be located) by comparing baseline characteristics between those retained and those who were lost to follow-up within each arm. We also examined absolute weight while adjusting for absolute weight at baseline as this approach has been shown to provide more precise estimates than percentage weight loss [37]. Because of the inclusion of participants in different diabetes categories, we also conducted analyses of the effect of intervention on weight within diabetes category by adding diabetes category and diabetes category by condition interactions to the model. For significant effects, we conducted tipping point analyses to judge the effect of missing data.

Results

Between February 2018 and March 2019, we screened 782 potential participants in 28 clusters. Of these, 494 individuals were eligible and were enrolled (Fig 1). Main reasons for ineligibility were BMI too low (56% of those ineligible) and blood pressure too high (29% of ineligibles). At the follow-up, completed between September 2018 and December 2019, 438 participants (28 clusters) were assessed including 215 (90%) in Lifestyle Africa and 223 (88%) in usual care.

Table 1 shows the baseline characteristics of participants by arm, while Table 2 shows baseline and follow-up levels of biometric measurements and medication use by arm. Participants were majority women and had a mean age of 68 years. Baseline characteristics were similar across groups, with 8% of participants in both groups employed and 31% of participants in both groups in the lowest income category and 57% in the middle category, although educational attainment was somewhat higher in the usual care group (52% advancing beyond eighth grade versus 44% in Lifestyle Africa). Mean BMI was in the obese range for both groups, although the Lifestyle Africa group had a lower baseline BMI (33.8 versus 35.2 in usual care). Although baseline HbA1c was similar between groups, more participants in the usual care group were in the prediabetic range (53.1% versus 44.2%), while fewer were in the diabetic range (22.4% versus 28.8%). Comparison of baseline characteristics between those who completed the trial and those lost to follow-up within each arm revealed notable differences in weight, BMI, HbA1c, and sex within the Lifestyle Africa arm. At baseline, mean weight was 91.89 kg for those lost to follow-up and 82.95 kg for those who completed the study [F(1,464)]= 6.10, p = 0.01]. Baseline BMI was 36.04 for those lost to follow-up and 33.54 for those who completed the study [F(1,464) = 3.10, p = 0.08]. Baseline HbA1c was 5.90 for those who were lost to follow-up and 6.41 for those who completed the study [F(1,464) = 4.40, p = 0.04]. With



Fig 1. CONSORT flow diagram. *Participants could be ineligible for more than one reason. BMI, body mass index; BP, blood pressure; HbA1c, hemoglobin A1c; Meds, medications.

https://doi.org/10.1371/journal.pmed.1003964.g001

respect to sex, males represented a higher proportion among those lost to follow-up than those who completed the study [24% versus 10%, respectively; F(1,464) = 4.25, p = 0.04]. The only notable difference within the usual care arm was with respect to sex, with a higher proportion of those lost to follow-up being males [19% versus 10%; F(1,464) = 2.45, p = 0.12].

The modal number of sessions held by CHWs across clubs was 17 (range of 10 to 17). Mean attendance across all sessions varied by cluster, ranging from 44% to 83%, with a mean of 61%. Across all clusters, 42% of participants attended at least 75% of the sessions held. Fidelity ratings based on observation of 74% of sessions indicated that mean completion rates of session facilitation elements (i.e., fully completed versus partially or not completed) for each club ranged from 62% to 95% (mean = 85.4%). Mean ratings on a 7-point Likert scale (1 = poor/never and 7 = excellent/always) of group facilitation skills, MI skills, and overall performance were 5.5 (range = 4.9 to 6.3), 5.4 (range = 4.7 to 6.2), and 5.5 (range = 4.7 to 6.3), respectively. There were no unanticipated problems and serious adverse events were rare, unrelated to the intervention, and similar across groups (11 in Lifestyle Africa versus 13 in usual care).

	Lifestyle Africa ($n = 240$)	Usual care $(n = 254)$
Number of clusters	15	13
Cluster size	16.00 (6.50)	19.54 (4.50)
Cluster size range	4 to 29	11 to 30
Sex		
Male	28 (11.7%)	28 (11.0%)
Female	212 (88.3%)	226 (89.0%)
Age (years)	67.6 (10.69)	67.8 (10.76)
Education attainment		
Below eighth grade	135 (56.5%)	118 (47.8%)
Above eighth grade	104 (43.5%)	129 (52.2%)
Relationship status		
Single	60 (25.0%)	72 (29.1%)
Married/living together	57 (23.8%)	59 (23.9%)
Widowed/divorced/separated	123 (51.2%)	116 (47.0%)
Employment status		
Unemployed/home/scholar	88 (36.7%)	76 (30.9%)
Employed	19 (7.9%)	20 (8.1%)
Pensioner	133 (55.4%)	150 (61.0%)
Income (monthly)		
Less than R1,600 (<\$126.10)	74 (30.8%)	75 (30.5%)
R1,600 to R3,200 (\$126.10 to \$252.20)	138 (57.5%)	140 (56.9%)
More than R3,200 (>\$252.20)	28 (11.7%)	31 (12.6%)
Diabetes medication*		
Metformin	48 (20.2%)	50 (20.1%)
Glimepiride	24 (10.1%)	25 (10.0%)
Insulin	6 (2.5%)	7 (2.8%)
BMI	33.82 (6.24)	35.22 (7.13)

Table 1. Participant baseline characteristics.

Data are n (%) or mean (SD). Rand to US Dollar exchange rate based on the mean of Feb 2018 and Feb 2019 rates. Counts do not add up to the group ns in all cases due to missing data; percentages are based on observed ns for that item.

*Diabetes medication categories are not mutually exclusive, i.e., some participants reported taking more than one. BMI, body mass index.

https://doi.org/10.1371/journal.pmed.1003964.t001

Empty models for the intent-to-treat analyses revealed club-level ICCs that ranged from 0.01 to 0.05 and repeated measures ICCs within person that ranged from 0.39 for systolic blood pressure to 0.98 for weight. The intent-to-treat analyses revealed the primary outcome of weight did not differentially change over time across conditions [time by condition interaction effect of F(1, 897) = 0.64, p = 0.42]. However, HbA1c did show differential change over time across conditions [time by condition interaction of F(1, 897) = 8.59, p = 0.004] with a Cohen's *d* effect size of 0.21. None of the other secondary outcomes showed differential change over time: systolic blood pressure, F(1, 896) = 0.82, p = 0.37; diastolic blood pressure, F(1, 896) = 0.42, p = 0.52.; triglycerides, F(1, 846) = 0.17, p = 0.68; or LDL, F(1, 839) = 0.28, p = 0.60.

For the analysis with outcomes at the end of year 1 as the dependent variable, club level ICCs were 0.05 or less, although they were slightly higher at 0.07 and 0.09 for systolic and diastolic blood pressure, respectively. Table 3 provides the parameter estimates from the models that included baseline values for outcomes (except for percent weight change) and study arm. Means (adjusted for the baseline value of the variable and clustering) for each treatment arm, the estimate of the mean difference between groups, and the 95% confidence interval (CI) for each are presented. Mean weight change was -0.61% (95% CI = -1.22, -0.01) in Lifestyle Africa and -0.44% (95% CI = -1.06, 0.18) in usual care, with the between-group model-

Table 2. Raw data for biometrics and medication use at baseline and follow-up.

	Lifest	yle Africa	Usual care			
	Baseline (<i>n</i> = 240)	Follow-up (<i>n</i> = 215)	Baseline (<i>n</i> = 254)	Follow-up (<i>n</i> = 223)		
Weight (kg)	83.94	82.36	86.55	86.05		
	(16.54)	(15.90)	(17.79)	(18.28)		
	240	215	254	223		
% Change in body weight	-	-0.66 (4.10) 215	-	-0.44 (3.61) 223		
Blood pressure (mm Hg) [†]						
Systolic	126.15	129.51	126.18	131.42		
	(15.82)	(19.89)	(15.59)	(21.17)		
	240	214	254	223		
	74.18	74.29	73.65	74.48		
	(9.51)	(11.03)	(9.37)	(10.77)		
	240	214	254	223		
LDL (mmol/L)	2.24	2.19	2.38	2.39		
	(0.83)	(0.85)	(0.83)	(0.84)		
	231	209	212	217		
Triglycerides (mmol/L)	1.73	1.73	1.70	1.73		
	(0.94)	(0.89)	(0.88)	(0.94)		
	232	212	215	222		
HbA1c (%)	6.37	6.32	6.27	6.38		
	(1.22)	(1.34)	(1.07)	(1.54)		
	240	212	254	223		
HbA1c category (<i>n</i>)	65	72	62	64		
Normal (<5.7%)	(27.1%)	(34.0%)	(24.4%)	(28.7%)		
Prediabetes (5.7% to 6.4%)	106	80	135	106		
	(44.2%)	(37.7%)	(53.1%)	(47.5%)		
Diabetes (>6.4%)	69	60	57	53		
	(28.8%)	(28.3%)	(22.4%)	(23.8%)		
BMI	33.82	33.31	35.22	35.09		
	(6.24)	(5.88)	(7.14)	(7.18)		
	240	215	254	223		
Diabetes medication $(n)^{\dagger\dagger}$	52 (21.8%)	36 (16.7%)	55 (22.1%)	41 (18.4%)		

Data are mean (SD) and n, except for diabetes category and medications, which are n (%).

[†]Some follow-up biometric data are missing due to occasional difficulties taking accurate measurements (device failures caused by ambient temperature or other unknown causes, injury preventing proper application of blood pressure cuff, etc.).

⁺⁺Diabetes medication use is based on reporting taking any one or a combination of metformin, insulin, or glimepiride. Baseline percentages reflect 7 participants (2 Lifestyle Africa and 5 usual care) not included in denominator due to missing data.

BMI, body mass index; HbA1c, Hemoglobin A1c; LDL, low-density lipoprotein.

https://doi.org/10.1371/journal.pmed.1003964.t002

adjusted difference not statistically significant (= -0.17%; 95% CI -1.04, 0.71; p = 0.71). Absolute mean weight at follow-up was 84.13 kg (95% CI 83.61, 84.65) in Lifestyle Africa and 84.40 kg (95% CI 83.88, 84.91) in usual care, with the between-group model-adjusted difference not statistically significant (= -0.27; 95% CI -1.00, 0.47). There were also no significant differences between treatment arms in blood pressure, LDL cholesterol, or triglycerides. However, there was a significant difference in HbA1c at follow-up between arms (mean difference = -0.24%, Hedges' g = 0.23, p = 0.001), with Lifestyle Africa participants averaging 6.23% (95% CI = 6.12, 6.34) and usual care averaging 6.47% (95% CI = 6.37, 6.58). Adding baseline BMI as a covariate to the models did not change the significance of the effect of treatment arm for any of the outcomes. We also considered whether HbA1c improvements might have been due to increases in medication use. However, results were essentially unchanged after removing from the

	ICC Lifestyle Africa Usual care Group difference		Group difference	SE	F Test	p	
Basic models-No added co	variates	· · ·		·			
% Weight change	0.026	-0.61 (-1.22, 0.01)	-0.44 (-1.06, 0.18)	-0.17 (-1.04, 0.71)	0.44	F(1,410) = 0.14	0.71
Weight	0.014	84.13 (83.61, 84.65)	84.40 (83.88, 84.91)	-0.27 (-1.00, 0.47)	0.37	F(1,409) = 0.51	0.48
Systolic BP (mm Hg)	0.097	129.56 (125.68, 133.45)	130.92 (126.94, 134.91)	-1.36 (-6.92, 4.21)	2.83	$\begin{array}{c c} 2.83 & F(1,408) = 0.23 \\ \hline \\ \hline \\ 1.40 & F(1,408) = 0.01 \\ \hline \end{array}$	
Diastolic BP (mm Hg)	0.062	74.23 (72.31, 76.16)	74.39 (72.42, 76.36)	-0.15 (-2.91, 2.60)	1.40	F(1,408) = 0.01	0.91
LDL (mmol/L)	0.022	2.27 (2.18, 2.35)	2.34 (2.25, 2.42)	-0.07 (-0.19, 0.05)	0.06	F(1,358) = 1.29	0.26
Triglycerides (mmol/L)	0.004	1.75 (1.63, 1.88)	1.78 (1.65, 1.90)	-0.02 (-0.20, 0.16)	0.09	F(1,369) = 0.07	0.80
HbA1c (%)	0.044	6.23 (6.12, 6.34)	6.47 (6.37, 6.58)	-0.24 (-0.39, -0.09)	0.08	0.08 F(1,406) = 10.32	
BMI as added covariate							
% Weight change	-	-0.65 (-1.27, -0.04)	-0.40 (-1.01, 0.21)	-0.26 (-1.13, 0.62)	0.44	0.44 F(1,409) = 0.33	
Weight	-	84.13 (83.61, 84.64)	84.40 (83.88, 84.92)	-0.27 (-1.01, 0.46)	0.37	F(1,408) = 0.54	0.46
Systolic BP (mm Hg)	-	129.49 (125.58, 133.39)	130.99 (126.98, 134.99)	-1.50 (-7.10, 4.10)	2.85	F(1,407) = 0.28	0.60
Diastolic BP (mm Hg)	-	74.10 (72.15, 76.06)	74.50 (72.50, 76.50)	-0.39 (-3.20, 2.41)	1.43	F(1,407) = 0.08	0.78
LDL (mmol/L)	-	2.27 (2.19, 2.35)	2.33 (2.24, 2.42)	-0.06 (-0.18, 0.06)	0.06	F(1,357) = 1.01	0.32
Triglycerides (mmol/L)	-	1.75 (1.62, 1.87)	1.78 (1.65, 1.91)	-0.03 (-0.21, 0.15)	0.09	F(1,368) = 0.31	0.72
HbA1c (%)	-	6.23 (6.12, 6.33)	6.47 (6.37, 6.58)	-0.25 (-0.40, -0.10)	0.08	$0.08 \qquad F(1,405) = 10.52$	
Excluding participants on d	iabetes medica	ation with reductions in	h HbA1c (n = 48)				
HbA1c (%)	-	6.20 (6.11, 6.30)	6.37 (6.28, 6.45)	-0.16 (-0.29, -0.03)	0.07	F(1,358) = 5.86	0.02

Table 3. ANCOVA model summaries for each outcome.

Means and (95% CIs). Means are adjusted means controlling for baseline.

BMI, body mass index; BP, blood pressure; CI, confidence interval; HbA1c, hemoglobin A1c; ICC, intraclass correlation coefficient; LDL, low-density lipoprotein; SE, standard error of the difference.

https://doi.org/10.1371/journal.pmed.1003964.t003

analysis the 48 participants who had a reduction in HbA1c and were on diabetes medication (see Table 3). A tipping point analyses was used to examine the impact of missing data at year 1 follow-up on HbA1c effects. Results indicated that the effect of treatment on HbA1c would still be significant (p = 0.04) if Lifestyle Africa participants with missing data at the year one follow-up had an increase in HbA1c that was 6 times that of the mean increase in the control group (i.e., an increase of 0.78%) with a resulting usual care mean of 6.45% and a Lifestyle Africa mean of 6.31%.

Analyses of effects within sex revealed no significant interaction effects for any of the outcomes (see Table 4). Treatment effects within sex indicated a slightly lower effect of treatment on HbA1c for men [group difference = 0.14%, 95% CI -0.33, 0.62; t(404) = 0.60, Hedges' g = 0.11, p = 0.55] than for women [group difference = 0.25%, 95% CI 0.10, 0.41; t(404) = 3.33, Hedges' g = 0.25, p = 0.002]. Analysis of effects on weight change within diabetes categories revealed no significant interaction effect [F(2, 405) = 0.49, p = 0.62].

	Male*			Female						
	Group difference	SE	t Test	p	Group difference	SE	t Test	p	Interaction test	p
% Weight loss	0.11 (-2.21, 2.44)	1.18	t(408) = 0.10	0.92	0.17 (-0.73, 1.08)	0.46	t(408) = 0.38	0.71	F(1, 408) = 0.01	0.96
Systolic BP (mm Hg)	4.26 (-7.15, 15.64)	5.80	t(406) = 0.73	0.46	1.04 (-4.61, 6.70)	2.88	t(406) = 0.36	0.72	F(1, 406) = 0.32	0.57
Diastolic BP (mm Hg)	1.53 (-4.37, 7.42)	3.00	t(406) = 0.51	0.61	0.004 (-2.81, 2.82)	1.43	t(406) = 0.01	0.99	F(1, 406) = 0.27	0.61
LDL (mmol/L)	-0.07 (-0.46, 0.31)	0.20	t(356) = 0.36	0.72	0.09 (-0.04, 0.21)	0.07	t(356) = 1.29	0.20	F(1, 356) = 0.58	0.45
Triglycerides (mmol/L)	0.21 (-0.28, 0.69)	0.25	t(367) = 0.84	0.40	0.003 (-0.18, 0.19)	0.09	t(367) = 0.03	0.97	<i>F</i> (1, 367) = 0.64	0.43
HbA1c (%)	0.14 (-0.33, 0.62)	0.24	t(404) = 0.60	0.55	0.25 (0.10, 0.41)	0.08	t(404) = 3.18	0.002	F(1, 404) = 0.19	0.67

Table 4. ANCOVA models separated by sex.

Means and (95% CIs). Parameter estimates are from a single model for each outcome where the F test is for the sex by treatment group interaction and the *t* test is for the differences between arms within each sex.

*Subgroup *n*'s for % weight loss are 22 (male control), 22 (male treatment), 201 (female control), and 193 (female treatment). Subgroup *n*'s for Systolic and Diastolic BP are 22 (male control), 21 (male treatment), 201 (female control), and 191 (female treatment). Subgroup *n*'s for LDL are 18 (male control), 20 (male treatment), 166 (female control), and 183 (female treatment). Subgroup *n*'s for triglycerides are 19 (male control), 21 (male treatment), and 185 (female treatment). Subgroup *n*'s for HbA1c are 22 (male control), 21 (male treatment), 201 (female control), and 191 (female control), and 191 (female treatment).

BP, blood pressure; CI, confidence interval; HbA1c, hemoglobin A1c; LDL, low-density lipoprotein; SE, standard error of the difference.

https://doi.org/10.1371/journal.pmed.1003964.t004

To analyze the clinical meaningfulness of the HbA1c effect, we categorized HbA1c values into normal (<5.7%), prediabetic (5.7% to 6.4%), and diabetic (>6.4%) categories and determined if improvements or deteriorations in diabetes category differed by group. In the control arm, 8% (4/49) of participants improved from the diabetes category and 13% (16/119) improved from the prediabetes category. In the Lifestyle Africa arm, 14% (9/65) of participants improved in the diabetes category, and 26% (24/93) improved in the prediabetes category. In the control arm, 13% (7/55) of participants deteriorated from the normal category, and 6% (7/119) deteriorated from the prediabetes category, whereas in the Lifestyle Africa arm 11% (6 of 54) deteriorated from the normal category, and 4% (4/93) deteriorated from the prediabetes category and a lower probability of lowering (improving) their HbA1c by at least 1 category and a lower probability of raising their diabetes category (odds ratio [OR] = 1.52, 95% CI 1.04, 2.22, p = 0.03).

Discussion

The Lifestyle Africa intervention was generally successfully delivered, and more than 40% of participants attended 75% or more of the sessions. While the intervention did not have a significant effect on the primary outcome of weight loss or the secondary indicators of blood pressure, LDL cholesterol, and triglycerides, the intervention had a significant positive effect on HbA1c. There was a mean decline in the intervention group and a mean increase in the control group with a mean HbA1c difference of 0.24%. This is approximately a quarter of the effect observed for metformin in randomized trials [38]. The effect was present for men and women, although the small number of men in the study hinders firm conclusions for men. Analysis using HbA1c-defined clinical diabetes categories indicated that the intervention group had nearly twice the odds of improving their diabetes category (i.e., moving from diabetes to prediabetes or prediabetes to normal).

Results are fairly consistent with the 2 prior studies conducted in LMIC countries that used a peer support model of treatment delivery and found some positive intervention effects [16,17]. The Kerala DPP had outcomes that consistently favored the intervention arm and the strongest effects on systolic blood pressure, triglycerides, and HbA1c, although none of these effects were statistically significant [16]. The study in Grenada [17] reported somewhat positive effects based on a nonlaboratory-based cardiovascular health index that included blood pressure and weight. In the present study, outcomes also consistently favored Lifestyle Africa, but only HbA1c effects were statistically significant. To our knowledge, our study is the first of its kind to show a significant effect on a biologic outcome, i.e., HbA1c.

Considering the positive effect of Lifestyle Africa on HbA1c, the lack of an effect on the primary outcome of weight loss was unexpected as blood glucose changes tend to be associated with weight loss [39–42]. However, our result is similar to the Kerala DPP study, which also found that the effect on HbA1c was stronger than on weight [16]. One explanation is that HbA1c changes may be related to dietary changes or changes in physical activity independent of weight loss [43,44].

With respect to intervention feasibility, results indicated that CHWs were able to deliver the sessions and that participants attended at acceptable levels. In almost all clubs, 16 or 17 sessions were held. Mean attendance was 61% across all sessions and all clubs, and 42% of participants attended at least 75% of the sessions. This is very consistent with delivery of 96% of planned sessions and median attendance of 60% observed in the peer-led Kerala DPP study [16]. These levels of attendance in resource-constrained environments compare well with attendance levels observed for the DPP in the US where less than 50% of participants attend all 16 core sessions [45]. Taken together with the quite high ratings of fidelity to the intervention, our findings support the feasibility of the Lifestyle Africa approach.

Although our results are similar to the prior studies utilizing nonprofessional interventionists in middle-income countries, the present study differs in some important ways. First, participants were mostly female, older rather than middle-aged, and mostly unemployed or receiving a pension rather than working. Following and engaging in a novel program and making changes in long-established behavior patterns may be more difficult for older participants who are experiencing both physical and cognitive decline. In addition, weight loss might have been hampered by a cultural norm of obesity among women in this population [46]. Second, the program was delivered entirely by lower resource CHWs, whereas the Kerala DPP utilized research program staff [16], a local resource provider, and expert panel members comprised of specialist advisors on diabetes, nutrition, and physical activity who provided education during some of the sessions. Use of video as a tool to provide all expert content and guide session flow in Lifestyle Africa may be particularly valuable in settings where use of these additional high resource personnel is not feasible. Video-based programs can also be relatively easily adapted to new languages, cultures, and contexts.

Strengths of the study include a rigorous cluster randomized design, a good level of participant attendance at sessions, and high levels of participant follow-up. These strengths are particularly notable in light of the daily challenges CHWs and participants face in this lowresource setting. Limitations of the study include the lack of a rigorous dietary intake measure, such as a 24-hour food recall, which could have identified any changes in dietary patterns that may have been related to HbA1c changes. Similarly, rigorous measures of physical activity and diabetes medication adherence would have been desirable. In addition, field-based measurements may include more error than clinical- and laboratory-based measures. Lipids were also assessed without fasting as this was judged to be too burdensome for participants who might have been experiencing food insecurity. Mean follow-up times were 7.5 to 14 days earlier for control participants than for Lifestyle Africa participants. This difference is unlikely to have had an impact on the observed HbA1c difference between groups as HbA1c reflects mean blood sugar levels over the prior 3 months. Although loss to follow-up was low overall, there was some indication that male sex, higher weight, and lower HbA1c were associated with a greater likelihood of dropping out of the intervention. Findings based on our sample of predominantly older, retired females of lower socioeconomic status should be generalized cautiously to other demographic groups and settings.

The results of this study, taken together with other early trials in LMICs are mixed, and thus how we proceed moving forward with these types of interventions are not clear-cut and may depend on what lens is used. On the one hand, both the current study as well as the Kerala study in India [16] did not achieve clinically significant weight loss, and the impact on HbA1c in our study was clinically modest, with no other significant outcomes in these 2 studies. These modest outcomes raise questions about the utility of continuing these lifestyle intervention adaptions in LMICs given that they can be costly and time consuming.

On the other hand, this study, the Kerala study [16], and the Grenada study [17] were nonetheless shown to be feasible with regard to use of local lay interventionists, recruiting and retaining study participants, and intervention engagement. While intervention development costs are significant, these instances incorporated several cost efficiencies such as use of lay interventions and group delivery of the intervention. Rather than jettisoning this approach entirely, perhaps improvement in intervention content and strategy, enhancing the dose of treatment, and other intervention delivery factors could lead to significant impact using this overall paradigm. For example, increasing "hands-on" activities such as physical activity and food preparation to allow practice of behavioral skills in session and thereby reduce complete reliance on self-regulation outside of sessions may be helpful. With regard to enhancing dose, longer interventions that provide ongoing support for necessary lifestyle changes (e.g., access and exposure to fresh fruits and vegetables) in settings where environmental barriers (e.g., food insecurity) are great may be worth testing.

In conclusion, results from this study indicate that the Lifestyle Africa intervention is feasible and had a clinically modest impact on HbA1c. The video-based design that avoids the need for expert involvement is highly scalable and could also be delivered via smart phone or an online platform. Before the intervention can be recommended for widespread scale-up, further studies are needed to confirm the feasibility of the intervention and generalizability of the findings to other countries and settings. In addition, cost-effectiveness analyses are important to show the feasibility and value of scale-up in LMICs.

Supporting information

S1 Checklist. CONSORT Checklist. CONSORT, Consolidated Standards of Reporting Trials. (DOCX)

Acknowledgments

The authors thank our community advisory board, our partner NGO organizations, the community health workers, and participants who made this study possible. We also thank Andrew T. Fox, PhD, for assistance in preparing the manuscript.

Disclaimers

The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health.

Author Contributions

Conceptualization: Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile, Ken Resnicow, Kathy Goggin.

Data curation: Kandace K. Fleming.

- Formal analysis: Kandace K. Fleming.
- **Funding acquisition:** Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile, Ken Resnicow, Kathy Goggin.
- **Investigation:** Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile, Ken Resnicow, Kandace K. Fleming, Emily A. Hurley, Joshua M. Smyth, Frank T. Materia, Kathy Goggin.
- Methodology: Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile, Ken Resnicow, Kandace K. Fleming, Emily A. Hurley, Joshua M. Smyth, Frank T. Materia, Estelle V. Lambert, Mara Z. Vitolins, Naomi S. Levitt, Kathy Goggin.
- Project administration: Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile, Kathy Goggin.
- Resources: Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile, Joshua M. Smyth.
- Supervision: Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile.
- Validation: Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile, Kandace K. Fleming.
- Writing original draft: Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile, Ken Resnicow, Kathy Goggin.
- Writing review & editing: Delwyn Catley, Thandi Puoane, Lungiswa Tsolekile, Ken Resnicow, Kandace K. Fleming, Emily A. Hurley, Joshua M. Smyth, Frank T. Materia, Estelle V. Lambert, Mara Z. Vitolins, Naomi S. Levitt, Kathy Goggin.

References

- 1. Noncommunicable diseases fact sheet [article online]. 2021 [cited 2021 Jun 3]. Available from https://www.who.int/news-room/fact-sheets/detail/noncommunicable-diseases
- Ogurtsova K, Da Rocha Fernandes JD, Huang Y, Linnenkamp U, Guariguata L, Cho NH, et al. IDF Diabetes Atlas: global estimates for the prevalence of diabetes for 2015 and 2040. Diabetes Res Clin Pract. 2017; 128:40–50. https://doi.org/10.1016/j.diabres.2017.03.024 PMID: 28437734
- 3. Schutte AE. Urgency for South Africa to prioritise cardiovascular disease management. Lancet Glob Health. 2019; 7(2):e177–8. https://doi.org/10.1016/S2214-109X(18)30476-5 PMID: 30528532
- Pheiffer C, Pillay-van Wyk V, Turawa E, Levitt N, Kengne AP, Bradshaw D. Prevalence of type 2 diabetes in South Africa: a systematic review and meta-analysis. Int J Environ Res Public Health. 2021; 18 (11):5868. https://doi.org/10.3390/ijerph18115868 PMID: 34070714
- 5. Popkin BM. The nutrition transition and obesity in the developing world. J Nutr. 2001; 131:871S–3S. https://doi.org/10.1093/jn/131.3.871S PMID: 11238777
- Kruger HS, Venter CS, Vorster HH. Physical inactivity as a risk factor for cardiovascular disease in communities undergoing rural to urban transition: the THUSA study. Cardiovasc J S Afr. 2003; 14:16–23. PMID: 12621539
- Micklesfield LK, Lambert EV, Hume DJ, Chantler S, Pienaar PR, Dickie K, et al. Socio-cultural, environmental and behavioural determinants of obesity in black South African women. Cardiovasc J Afr. 2013; 24(9–10):369–75. https://doi.org/10.5830/CVJA-2013-069 PMID: 24051701
- Diabetes Prevention Program Research Group. 10-year follow-up of diabetes incidence and weight loss in the Diabetes Prevention Program Outcomes Study. Lancet. 2009; 374:1677–86. https://doi.org/ 10.1016/S0140-6736(09)61457-4 PMID: 19878986
- Katula JA, Vitolins MZ, Rosenberger EL, Blackwell CS, Morgan TM, Lawlor MS, et al. One-year results of a community-based translation of the Diabetes Prevention Program: healthy-living partnerships to prevent diabetes (HELP PD) project. Diabetes Care. 2011; 34:1451–7. https://doi.org/10.2337/dc10-2115 PMID: 21593290

- Look AHEAD Research Group. Reduction in weight and cardiovascular disease risk factors in individuals with type 2 diabetes: one-year results of the Look AHEAD trial. Diabetes Care. 2007; 30:1374–83. https://doi.org/10.2337/dc07-0048 PMID: 17363746
- Aekplakorn W, Tantayotai V, Numsangkul S, Tatsato N, Luckanajantachote P, Himathongkam T. Evaluation of a community-based diabetes prevention program in Thailand: a cluster randomized controlled trial. J Prim Care Community Health. 2019; 10:1–8.
- Assunção MCF, Gigante DP, Cardoso MA, Sartorelli DS, Santos IS. Randomized, controlled trial promotes physical activity and reduces consumption of sweets and sodium among overweight and obese adults. Nutr Res. 2010; 30:541–9. https://doi.org/10.1016/j.nutres.2010.07.006 PMID: 20851308
- Sun J, Wang Y, Chen X, Chen Y, Feng Y, Zhang X, et al. An integrated intervention program to control diabetes in overweight Chinese women and men with type 2 diabetes. Asia Pac J Clin Nutr. 2008; 17:514–24. PMID: 18818173
- Rawal LB, Tapp RJ, Williams ED, Chan C, Yasin S, Oldenburg B. Prevention of type 2 diabetes and its complications in developing countries: a review. Int J Behav Med. 2012; 19:121–33. https://doi.org/10. 1007/s12529-011-9162-9 PMID: 21590464
- Ruggiero L, Castillo A, Quinn L, Hochwert M. Translation of the diabetes prevention program's lifestyle intervention: role of community health workers. Curr Diab Rep. 2012; 12:127–37. https://doi.org/10. 1007/s11892-012-0254-y PMID: 22350806
- Thankappan KR, Sathish T, Tapp RJ, Shaw JE, Lotfaliany M, Wolfe R, et al. A peer-support lifestyle intervention for preventing type 2 diabetes in India: a cluster-randomized controlled trial of the Kerala Diabetes Prevention Program. PLoS Med. 2018; 15:e1002575. <u>https://doi.org/10.1371/journal.pmed.</u> 1002575 PMID: 29874236
- Latina J, Fernandez-Jimenez R, Bansilal S, Sartori S, Vedanthan R, Lewis M, et al. Grenada heart project-community health action to encourage healthy behaviors (GHP-CHANGE): a randomized control peer group-based lifestyle intervention. Am Heart J. 2020; 220:20–8. https://doi.org/10.1016/j.ahj.2019. 08.022 PMID: 31765932
- Murphy JP, Moolla A, Kgowedi S, Mongwenyana C, Mngadi S, Ngcobo N, et al. Community health worker models in South Africa: a qualitative study on policy implementation of the 2018/19 revised framework. Health Policy Plan. 2021; 36(4):384–96. <u>https://doi.org/10.1093/heapol/czaa172</u> PMID: 33367608
- Bender MS, Cooper BA, Flowers E, Ma R, Arai S. Filipinos fit and trim: a feasible and efficacious DPPbased intervention trial. Contemp Clin Trials Commun. 2018; 12:76–84. <u>https://doi.org/10.1016/j.</u> conctc.2018.09.004 PMID: 30294699
- Catley D, Puoane T, Tsolekile L, Resnicow K, Fleming K, Hurley EA, et al. Adapting the Diabetes Prevention Program for low and middle-income countries: protocol for a cluster randomised trial to evaluate 'Lifestyle Africa'. BMJ Open. 2019; 9:e031400. https://doi.org/10.1136/bmjopen-2019-031400 PMID: 31719084
- Catley D, Puoane T, Goggin K, Tsolekile LP, Resnicow K, Fleming K, et al. Adapting the Diabetes Prevention Program for low- and middle-income countries: preliminary implementation findings from Lifestyle Africa. Transl Behav Med. 2020; 10(1):46–54. <u>https://doi.org/10.1093/tbm/ibz187</u> PMID: 31909412
- 22. Statistics South Africa. Census 2011 statistical release. Pretoria, South Africa; 2012.
- 23. Malhotra R, Hoyo C, Østbye T, Hughes G, Schwartz D, Tsolekile L, et al. Determinants of obesity in an urban township of South Africa. South Afr J Clin Nutr 2008; 21:315–20.
- Look AHEAD Research Group. Baseline characteristics of the randomized cohort from the Look AHEAD (action for health in diabetes) research study. Diab Vasc Dis Res. 2006; 3:202–15. <u>https://doi.org/10.3132/dvdr.2006.031</u> PMID: 17160917
- Diabetes Prevention Program Research Group. Reduction in the incidence of type 2 diabetes with lifestyle intervention or metformin. N Engl J Med. 2002; 346:393–403. <u>https://doi.org/10.1056/</u> NEJMoa012512 PMID: 11832527
- Venditti EM, Kramer MK. Necessary components for lifestyle modification interventions to reduce diabetes risk. Curr Diab Rep. 2012; 12:138–46. https://doi.org/10.1007/s11892-012-0256-9 PMID: 22350807
- Puoane T, Bradley H, Hughes G. Community intervention for the emerging epidemic of non-communicable diseases. South Afr J Clin Nutr. 2006; 19:56–62.
- Puoane TR, Tsolekile L, Igumbor EU, Fourie JM. Experiences in developing and implementing health clubs to reduce hypertension risk among adults in a South African population in transition. Int J Hypertens. 2012; 2012:913960. https://doi.org/10.1155/2012/913960 PMID: 22957212

- Cole-Lewis H, Kershaw T. Text messaging as a tool for behavior change in disease prevention and management. Epidemiol Rev. 2010; 32:56–69. <u>https://doi.org/10.1093/epirev/mxq004</u> PMID: 20354039
- McMaster F, Resnicow K. Validation of the one pass measure for motivational interviewing competence. Patient Educ Couns. 2015; 98:499–505. https://doi.org/10.1016/j.pec.2014.12.014 PMID: 25624118
- **31.** Arabadjief M, Nichols JH. Evaluation of the Afinion AS100 point-of-care analyzer for hemoglobin A1c. Point Care. 2009; 8:11–5.
- Jain A, Rao N, Sharifi M, Bhatt N, Patel P, Nirmal D, et al. Evaluation of the point of care Afinion AS100 analyser in a community setting. Ann Clin Biochem. 2017; 54:331–41. https://doi.org/10.1177/ 0004563216661737 PMID: 27422136
- Pickering TG, Hall JE, Appel LJ, Falkner BE, Graves J, Hill MN, et al. Recommendations for blood pressure measurement in humans and experimental animals. Hypertension. 2005; 45:142–61. <u>https://doi.org/10.1161/01.HYP.0000150859.47929.8e PMID: 15611362</u>
- Harris PA, Taylor R, Thielke R, Payne J, Gonzalez N, Conde JG. Research electronic data capture (REDCap)—a metadata-driven methodology and workflow process for providing translational research informatics support. J Biomed Inform. 2009; 42:377–81. <u>https://doi.org/10.1016/j.jbi.2008.08.010</u> PMID: 18929686
- Ali MK, Echouffo-Tcheugui JB, Williamson DF. How effective were lifestyle interventions in real-world settings that were modeled on the Diabetes Prevention Program? Health Aff. 2012; 31:67–75. https:// doi.org/10.1377/hlthaff.2011.1009 PMID: 22232096
- Lindström J, Louheranta A, Mannelin M, Rastas M, Salminen V, Eriksson J, et al. The Finnish diabetes prevention study (DPS): lifestyle intervention and 3-year results on diet and physical activity. Diabetes Care. 2003; 26:3230–6. https://doi.org/10.2337/diacare.26.12.3230 PMID: 14633807
- Vickers AJ. The use of percentage change from baseline as an outcome in a controlled trial is statistically inefficient: a simulation study. BMC Med Res Methodol. 2001; 1(1):1–4.
- Hirst JA, Farmer AJ, Ali R, Roberts NW, Stevens RJ. Quantifying the effect of metformin treatment and dose on glycemic control. Diabetes Care. 2012; 35(2):446–54. https://doi.org/10.2337/dc11-1465 PMID: 22275444
- Ockene IS, Tellez TL, Rosal MC, Reed GW, Mordes J, Merriam PA, et al. Outcomes of a Latino community-based intervention for the prevention of diabetes: the Lawrence Latino Diabetes Prevention Project. Am J Public Health. 2012; 102:336–42. https://doi.org/10.2105/AJPH.2011.300357 PMID: 22390448
- 40. Fine A, Gallaway MS, Dukate A. Prevention in prison: the diabetes prevention program in a correctional setting. Diabetes Spectr. 2019; 32:331–7. https://doi.org/10.2337/ds18-0080 PMID: 31798291
- Sepah SC, Jiang L, Ellis RJ, McDermott K, Peters AL. Engagement and outcomes in a digital Diabetes Prevention Program: 3-year update. BMJ Open Diabetes Res Care. 2017; 5:e000422. <u>https://doi.org/ 10.1136/bmjdrc-2017-000422</u> PMID: 28948027
- 42. Toro-Ramos T, Michaelides A, Anton M, Karim Z, Kang-Oh L, Argyrou C, et al. Mobile delivery of the Diabetes Prevention Program in people with prediabetes: randomized controlled trial. JMIR Mhealth Uhealth. 2020; 8:e17842. https://doi.org/10.2196/17842 PMID: 32459631
- 43. Shah SZ, Karam JA, Zeb A, Ullah R, Shah A, Haq IU, et al. Movement is improvement: the therapeutic effects of exercise and general physical activity on glycemic control in patients with type 2 diabetes mellitus: a systematic review and meta-analysis of randomized controlled trials. Diabetes Ther. 2021; 12:707–32. https://doi.org/10.1007/s13300-021-01005-1 PMID: 33547579
- Wali JA, Solon-Biet SM, Freire T, Brandon AE. Macronutrient determinants of obesity, insulin resistance and metabolic health. Biology. 2021; 10:336. <u>https://doi.org/10.3390/biology10040336</u> PMID: 33923531
- Ely EK, Gruss SM, Luman ET, Gregg EW, Ali MK, Nhim K, et al. A national effort to prevent type 2 diabetes: participant-level evaluation of CDC's national diabetes prevention program. Diabetes Care. 2017; 40:1331–41. https://doi.org/10.2337/dc16-2099 PMID: 28500215
- 46. Puoane T, Steyn K, Bradshaw D, Laubscher R, Fourie J, Lambert V, et al. Obesity in South Africa: the South African demographic and health survey. Obes Res. 2002; 10(10):1038–48. https://doi.org/10. 1038/oby.2002.141 PMID: 12376585